Author's response to reviews

Title: Multiplex SNaPshot for detection of BRCA1/2 common mutations in Spanish and Spanish related breast/ovarian cancer families

Authors:

Sandra Filippini (sandraefilip@yahoo.com.ar)
Ana Blanco (anablancop@hotmail.com)
Ana Fernandez-Marmiesse (ana.fernandez.marmiesse@sergas.es)
Vanesa Alvarez-Iglesias (vaneiml@usc.es)
Clara Ruiz-Ponte (crponte@usc.es)
Angel Carracedo (apimlang@usc.es)
Ana Vega (apimlapv@usc.es)

Version: 3 Date: 21 June 2007

Author's response to reviews:

MS: 5563644021342358
Journal: BMC Medical Genetics
Multiplex SNaPshot for detection of BRCA1/2 founder mutations in Spanish and Spanish related breast/ovarian cancer families. Sandra Filippini, Ana Blanco, Ana Fernandez-Marmiesse, Vanesa Alvarez-Iglesias, Clara Ruiz-Ponte, Angel Carracedo and Ana Vega

Dear Dr. Liz Hoffman

We are sending the final version of our manuscript with all the formatting changes requested done:

We had structured the Abstract into Background, Methods, Results, and Conclusions.

We had removed the Keywords.

We had rename Materials & Methods to 'Methods'.

We had removed Website addresses from the text and cited from the main reference list only.

We made the words 'Figure legends' the section heading.

Whit best wishes

Ana Vega, PhD
Fundacion Publica Galega de Medicina Xenomica-SERGAS
Edificio de Consultas, Planta -2
Hospital Clinico de Santiago
Santiago de Compostela, 15706, Spain
Tel: 981-951490/1
Fax: 981-951473
E-mail: apimlapv@usc.es